Author's response to reviews

Title: Potent alpha-glucosidase and alpha-amylase inhibitory activities of standardized 50% ethanolic extracts and sinensetin from Orthosiphon stamineus Benth as anti-diabetic mechanism

Authors:

Elsnoussi Ali Hussin Mohamed Dr (aliali692002@yahoo.com)
Mun Fei Yam Dr (yammunfei@yahoo.com)
Mohamed Jamshed Ahmad Siddiqol Dr (siddiquimj@gmail.com)
Mohd. Zaini Asmawi Prof (amzaini@usm.my)

Version: 2 Date: 24 August 2012

Author's response to reviews: see over
Covering letter

To:
Editors-in-Chief:
BMC Complementary and Alternative Medicine

Object: Submission of manuscript

Dear Editor-in-chief

Herewith, we are submitting a manuscript named “Potent alpha-glucosidase and alpha-amylase inhibitory activities of standardized 50% ethanolic extracts and sinensetin from Orthosiphon stamineus Benth as anti-diabetic mechanism” for publication in your esteemed journal. The manuscript is an original work and is not under consideration for publication elsewhere. The materials of the manuscript will not be submitted for publication elsewhere so long as it is under consideration by the BMC Complementary and Alternative Medicine. Rats used in study have been treated according to guidelines of the Biological Research Ethics Committee of the School of Pharmaceutical Sciences at University Sains Malaysia. All authors have contributed to this work. They have read the manuscript, agreed the work is ready for submission to BMC Complementary and Alternative Medicine and accepted responsibility for the manuscript’s contents. All authors have declared that they have no actual or potential competing financial interest regarding the submitted manuscript.

Regards,

Mun Fei Yam, PhD
Senior Lecturer
School of Pharmaceutical Sciences,
Universiti Sains Malaysia, 11800 Penang, Malaysia
Tel:+604-6534865
Fax: +604-6570017
E-mail: yammunfei@yahoo.com